\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage[flushleft]{threeparttablex}

\begin{document}


\def\RunningAuthor{Dewi Melani Hariyadi et al.}
\firstPage{611}
\articleType{Original Article}
\receivedDate{20.09.2019}
\acceptedDate{18.12.2019}
\revisedDate{12.12.2019}
\journalVolume{2020, 11}
\journalIssue{1}
\journalDoi{ijrps.v11i1.1864}
\copyrightYear{2020}
\def\authorCount{5}
\def\affCount{4}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{\textbf{Characterization }\textbf{and antibacterial activity of }\textbf{\textit{cocos Nucifera L.} Meat extract }\textbf{and powder } \textbf{as} a \textbf{drug and} \textbf{cosmetic agent}}
\author{Dewi~Melani~Hariyadi\textsuperscript{*}\textsuperscript{1},  
            Sisunandar\textsuperscript{2},  
            Suciati\textsuperscript{3},  
            Isnaeni\textsuperscript{4}, 
            Noorma~Rosita\textsuperscript{1}~\\[5pt]{\textsuperscript{1}Pharmaceutics Department, Faculty of Pharmacy, Airlangga University,  Jl. Mulyorejo, Surabaya 60286, Indonesia}~\\{\textsuperscript{2}Biology Department, Universitas Muhammadiyah Purwokerto, Indonesia}~\\{\textsuperscript{3}Pharmacognosy and Phytochemistry Department, Faculty of Pharmacy, Airlangga University, Jl. Mulyorejo, Surabaya 60286, Indonesia}~\\{\textsuperscript{4}Pharmaceutical Chemistry Department, Faculty of Pharmacy, Airlangga University, Jl. Mulyorejo, Surabaya 60286, Indonesia}}

\begin{abstract}
Cocos nucifera L. or Kopyor coconut is the natural material plant that has nutrient content, including carbohydrates, proteins, fats and fatty acids. The potential of Kopyor coconut is mainly produced from water and soft flesh or meat. Some benefits of the meat have not been widely studied as an active drug agent. This study was aimed to identify and characterize kopyor meat extract and to test activity as an antibacterial agent. The maceration method was used to extract kopyor meat. The optimized extraction resulted in a yield of 23\% efficiency. Kopyor meat extract was identified in terms of loss on drying, total ash, acid-insoluble ash, extract content and saponification value. Evaluation of the antibacterial activity of both dried meat kopyor and extract were conducted. The standardized extract had a loss on drying of 35\%, total ash of 8.95\%, acid-insoluble ash of 31\%, and extract content soluble in water and ethanol of 56.9\% and 0.6\% respectively. The saponification value showed a value of 56. It was shown that both powder and Cocos nucifera extract had as same high activity as an antibacterial agent against Staphylococcus aureus, Pseudomonas aeruginosa and Staphylococcus epidermidis therefore this recommended for further formulation and evaluation as drug and cosmetic.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Characterization,\newline Antibacterial activity,\newline Kopyor meat,\newline Extract
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Dewi~Melani~Hariyadi~\\ Phone:\ ~\\  Email:\ dewi-m-h@ff.unair.ac.id
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Cocos nucifera L. or coconut kopyor is a plant that almost all parts of the plant have been used by humans. Therefore, it is considered a versatile plant, especially for coastal communities and is a type of plant from the Arecaceae. The most widely used main component of coconut kopyor is young coconut water and soft flesh. Some of the benefits are kopyor coconut water for antipyretic and anti-inflammatory activities \unskip~\citep{660223:15546537,660223:15546539}. While the benefits of the fruit flesh themselves have not been widely studied as medicinal or cosmetic ingredients. The content of the main bigger components in the fruit flesh is hypothesized to produce pharmacological activity as a medicinal and cosmetic ingredient anti-oxidants, anti-osteoporosis, antidiabetic, antineoplastic, bactericidal, antihelminthic, antimalarial, leishmanicidal, antifungal, and antiviral activities\unskip~\citep{660223:15546532,660223:15546536,660223:15546533}. To be able to provide great benefits in developing its potential as a medicinal and cosmetic ingredient, Kopyor coconut extract needs to be extracted with the proper method, which needs to be identified and characterized by all the main components contained. The utilization was mostly still in water or oil. Extracts of Kopyor coconut were hypothesized and indicated for anti-itching, anti-bacterial, antioxidant, anti-viral, analgesic and anti-inflammatory drugs\unskip~\citep{660223:15546536}. While the benefits in cosmetics for skin and hair care can also reduce psoriasis, eczema, soften the skin, reduce skin dryness, prevent wrinkles and black spots as well as anti UV radiation \unskip~\citep{660223:15546535}. The extraction process is the first process before kopyor being tested for pre-formulation, formulation and further activity testing. Extraction is a process of extracting an active compound from material or simplicia using a suitable solvent. Extraction can be done by various methods, according to nature and purpose. This research will study the identification, characterization and antibacterial activity of kopyor coconut extract (Cocos nucifera L.).  
    
\section{Materials and Methods}
Cocos nucifera L. extract and dried kopyor, aquadest, methanol (p.a), hexane (p.a), HCl (p.a), NaOH (p.a), FeCl\ensuremath{_{3}} (p.a), filter paper (Whatman No. 1).

\textbf{Sample Preparation}

Samples of kopyor powder (Cocos nucifera Linn) were obtained from the Coconut Research Center in Purwokerto. Samples of this research were divided into kopyor powder and extract. 

\textbf{Extraction of \textit{Cocos nucifera L.} by maceration method using 70\% ethanol}

Kopyor powder samples (Cocos nucifera L) were weighed as much as 300 grams. Samples were put in a maceration container and 70\% ethanol containing 1\% HCl with a ratio of 1: 4 (w/v) was added until all submerged and tightly closed, followed by keeping for 24 hours and stirring several times during maceration. Samples were filtered using filter paper Whatman No. 1 and were separated. The filtrate was then macerated again with a new solution of 70\% ethanol containing 1\% HCl. This process was done three times with each time of 24 hours. The extract obtained was evaporated at 50\ensuremath{^\circ}C and a speed of 80 rpm until a thick extract was obtained. 

\textbf{Phytochemical Screening Analysis}

\textbf{Alkaloid Test}

The Alkaloid Test was carried out by the Dragendorff method. A sample of 3 ml was placed in a test tube and then 5 ml of methanol was added, followed by the addition of 3 ml of ammonia to a pH value of 8-9. The methanol extract was filtered and 2 ml of sulfuric acid was added and was shaken to get 2 layers. The top layer (sulfate) was taken 5 drops, then 1 drop of Dragendorf reagent was added and the formation of orange deposits showed the presence of alkaloids\unskip~\citep{660223:15546538}.

\textbf{Saponin Test}

The Saponin test was carried out using the Forth method by inserting 2 ml of the sample into a test tube and then 10 ml of distilled water was added. The mixture was heated for 2-3 minutes and was cooled down. After being shaken for 30 seconds, changes occurred. If the permanent solid foam was formed (permanently for 30 seconds), this indicated the presence of saponins \unskip~\citep{660223:15546538}.

\textbf{Flavonoid Test}

The flavonoid test was carried out using the Shinoda method. A total of 0.5 ml of the sample was dropped on a glass preparation. Next, 3 drops of methanol were added and were stirred until homogeneous. Following that, a small piece of Mg tape was added, then 3 drops of concentrated HCl were added. The formation of yellow, orange, red, or blue indicated the presence of flavonoid compounds.

\textbf{Phenolic Test}

A total of 0.50 ml of sample was dropped on a glass preparation, then 3 drops of methanol were added and were stirred until homogeneous, followed by the addition of 3 drops of 5\% FeCl\ensuremath{_{3}}. The formation of green, red, purple, or blue indicated the presence of phenolic compounds\unskip~\citep{660223:15546538}.

\textbf{Flavonoid Test using Thin Layer Chromatography (TLC)}

The filtrate on phytochemical screening was plated on silica gel 60 F254 plates, then it was rubbed with a mixture of butanol: acetic acid: water at ratio 3: 1: 1, then it was dried and was observed using 254 nm and 366 nm UV light. Furthermore, the plates were sprayed with ammonia, were dried and were re-observed with 254 nm and 366 nm UV light \unskip~\citep{660223:15546538}.\textit{\textbf{\space }}

\textbf{Anthocyanine Content Test}

This method was used to test the existence of anthocyanin. The first method was by heating with 2M HCl for 2 minutes at 100\ensuremath{^\circ}C, then the sample color was observed. If the red color in the sample did not change (steady), this showed the presence of anthocyanin. The second way was by adding samples with drops of 2M NaOH. If the red color turns blue-green and fades slowly, it was indicated anthocyanin \unskip~\citep{660223:15546534}.

\textbf{Characterization of Kopyor Extract}

\textbf{Moisture Content}

The moisture content of the extract was measured using the Moisture Content Analyzer after the drying process with Freeze Dryer.

\textit{\textbf{E}} \textit{\textbf{xtract }} \textbf{standardization } \textbf{and characterization}

\textbf{Determination of }\textbf{Ash} \textbf{ Levels}

2 grams of the refined test material was weighed, and then it was inserted into the silicate crucible, which has been glowed and anchored. Spread it slowly until the charcoal runs out in the furnish temperature of 800\ensuremath{^\circ}C. The crucible was then cooled and weighed. If the charcoal cannot be removed, enough hot water was added. Hot water was stirred and then filtered using ash-free filter paper. The filtrate was refined along with filter paper in the same silicate crucible in the 800\ensuremath{^\circ}C and it was furnished to a fixed weight. The total ash content was measured and was expressed in \% w/w.

\textbf{Determination of Non-Soluble Acid Levels}

The ash obtained in the determination of total ash content was boiled with 25.0 mL dilute HCl for 5 minutes. Insoluble parts were gathered in acid using ash-free filter paper. The filter paper was rinsed with enough hot water and spread the ash-free filter paper into the crucible in the furnish temperature of 800\ensuremath{^\circ}C to a fixed weight. The insoluble ash content in the acid was calculated against the weight of the tested samples and was expressed in \% w/w.

\textbf{Determination of Water-Soluble }\textbf{Extract } \textbf{Content}

5 grams of dried powder was weighed and inserted into Erlenmeyer. 100.0 mL of chloroform saturated water (water: chloroform = 1: 1) was added and was shaken many times for 6 hours. This mixture was then left for 18 hours and was filtered. The 20.0 mL of filtrate was filtered in a dish that has been anchored at 105\ensuremath{^\circ}C. Heating the filtrate into the oven at 105\ensuremath{^\circ}C until the weight remains and the level of the water-soluble extract was calculated in \%w/w.

\textbf{Determination of }\textbf{Ethanol }\textbf{Soluble }\textbf{Extract } \textbf{Content}

5 grams of dried powder was weighed and inserted into Erlenmeyer. 100.0 mL of concentrated ethanol pro analysis was added and was shaken many times for 6 hours. This mixture was then left for 18 hours and was filtered. The 20.0 mL of filtrate was filtered in a dish that has been anchored at 105\ensuremath{^\circ}C. Heating the filtrate into the oven at 105\ensuremath{^\circ}C until the weight remains and the level of the ethanol-soluble extract was calculated in \%w/w.

\textbf{Determination of }\textbf{loss on } \textbf{drying }

1 gram of extract was weighed in a weighing bottle that has been anchored and dried at 105\ensuremath{^\circ}C. The smooth extract was flattened by tapping the weighing bottle onto the floor. The weighing bottle was inserted into the oven at 105 \ensuremath{^\circ}C. Weight up to constant weights and was expressed in \% w/w 

\textbf{Saponification value}

2.5 grams of sample in Erlenmeyer 250 mL was weighed. 25.0 mL of KOH - ethanol 0.5 NLV was added. The mixture was continued to reflux for 30 minutes. Then 1 mL of the PP indicator was added. The excess of KOH was titrated using 0.5 N HCl. The saponification value was calculated. 

\textbf{\textbf{Characterization of fatty acid content}}

The identification of fatty acid content in kopyor extract by the GC-MS method was conducted. Methyl esters from fatty acids were made by dissolving samples in HCl (1.5 M, 15 mL) in methanol. The sample solution was then refluxed at 60\text{\textordmasculine}C for 2 hours using a water bath. Then toluene (1 mL) was added, the solvent was then evaporated in the rotary evaporator. The FAME reaction was purified with SiO2 eluted with hexane / EtOAc (1/1) to obtain FAME derivatives, which were then analyzed by GC-MS. 

The GC-MS analysis was performed on GC-FID with Agilent Technologies 6890N and GC-MSD with the Agilent 6973 series equipped with the Willey 7n.1 database in the HP-5 column (30 mx 0.250 mm x 0.25 \ensuremath{\mu}m). The temperature for GC-MS starts at 100\text{\textordmasculine}C, then increased to 250 \text{\textordmasculine} at a rate of 16\text{\textordmasculine}C/minute, and was held for 20 minutes.

\textbf{The total plate number of dried kopyor and kopyor extract}

A total of 25 grams of dried coconut was inserted into 225 mL of sterile saline solution, and it was shaken overnight on a shaking incubator on 30 \ensuremath{^\circ}C milk with a speed of 150 rpm. 1 mL of suspension was pipetted, 9 mL of saline solution was added and vortexed, then it was made diluted to 10\ensuremath{^{-4}}. From 1 mL was pipetted and was put into sterile petridisk, 10 mL \textit{nutrient} media was added until the temperature was 45\ensuremath{^{o}}C, then was incubated at 37\ensuremath{^{o}}C for 24-48 hours. The number of colonies was then calculated. 

\textbf{Anti}\textbf{bacterial} \textbf{ activity test}

For microbial test preparation, a representation of gram-negative and positive bacteria was used. Pure culture of test microbes was rejuvenated in nutrient agar sloping media. Setelang was incubated 24 hours at 37\ensuremath{^\circ}C plus 10 mL of saline solution, shaken with a vortex to release the culture from agar. 

The suspension or inoculum was measured by a spectrophotometer and was diluted to 25\% transmittance. A total of 5 \ensuremath{\mu}L of suspension was inoculated into 8 mL nutrient seed media at 45\ensuremath{^\circ}C. It was shaken and was poured over the surface of the nutrient base media, which had been compacted in sterile petridisk. The hole was made to use a perforator with a diameter of 0.8 cm and a height of 0.5 cm. Test media was ready for use. 

In the testing hole, 50 \ensuremath{\mu}L of the test solution both from the powder and kopyor extract was loaded and was dissolved in DMSO with a concentration of 50 mg/10 mL. For comparison, a 200 ppm standard solution was used. Incubation was carried out for 24 hours at 37\ensuremath{^\circ}C. The observed inhibition and zone diameter were measured (mm).
    
\section{Results and Discussion}
\textbf{Characterization of kopyor } \textbf{\textit{Cocos nucifera L.}}

The physical performance of the organoleptic characterization of fresh kopyor and dried kopyor coconut meat, as seen in Table~\ref{tw-d520eda6a06c} and Table~\ref{tw-65009cd39aaa}.


\begin{table}[!htbp]
\caption{\boldmath {Characteristics of Fresh Kopyor} }
\label{tw-d520eda6a06c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4948\linewidth-2\tabcolsep}p{\dimexpr.5052\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter  & Observation  \\
\tblmidrule 
Form &
  Soft\\
Color &
  White\\
Smell &
  Typical coconut\\
Taste &
  Sweet\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\begin{table}[!htbp]
\caption{\boldmath {Characteristics of dried kopyor resulting from a frozen, dried process} }
\label{tw-65009cd39aaa}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.436\linewidth-2\tabcolsep}p{\dimexpr.564\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter  & Observation  \\
\tblmidrule 
Form &
  Hard\\
Color &
  White yellowish\\
Smell &
  Typical coconut\\
Taste &
  Sweet\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}
\textbf{Extraction}

Simplicia of dried kopyor meat was extracted by the maceration method. A sample of 272 grams of dried kopyor was added by one Litre of 70\% ethanol and was soaked for 24 hours. The filtrate and the residue were separated using a Buchner funnel. The residue was then macerated again twice. All filtrates of extract were collected, the solvent was evaporated using a rotary evaporator until the remaining water phase and then was dried with freeze-drying. The extract was obtained was 62.35 grams and this was equal to extract yield of 22.9 \%.

\textbf{Characterization of kopyor coconut extract}

\textbf{Organoleptic characterization}

Kopyor coconut extract after maceration and drying with freeze dryer, the results were obtained and were then organoleptically characterized, as shown in Table~\ref{tw-d7c57cf9a43c}.


\begin{table}[!htbp]
\caption{\boldmath {Characteristics of the kopyor coconut extract} }
\label{tw-d7c57cf9a43c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4446\linewidth-2\tabcolsep}p{\dimexpr.5554\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter  & Observation  \\
\tblmidrule 
Form &
  Hard sticky\\
Color &
  Brownish\\
Smell &
  Typical coconut\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}
\textbf{Moisture Content}

Results of the moisture content of freeze-dried kopyor meat were 6.42\%. The extract did not extract because it was a thick extract and contained high amounts of oil.

\textbf{Standardization of extracts and characterization}

Results of standardization of extracts referred to the Indonesian Pharmacopoeia 5\ensuremath{^{th}} edition. Standardization was included a loss on drying, total ash content, acid insoluble ash content, water-soluble extract content and ethanol-soluble extract content obtained was shown in Table~\ref{tw-f430ff8f58e3}.


\begin{table}[!htbp]
\caption{\boldmath {Standardization of cocos Nucifera meat extract} }
\label{tw-f430ff8f58e3}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.5242\linewidth-2\tabcolsep}p{\dimexpr.4758\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Parameter of Standardization  & Measurement results (\% w/w) \\
\rowcolor{kwdboxcolor} & \cAlignHack Average \ensuremath{\pm} SD \\
\tblmidrule 
Loss on drying &
  34.96 \ensuremath{\pm} 0.30\\
Total ash content &
  8.95 \ensuremath{\pm} 0.10\\
Levels of acid-insoluble ash &
  30.94 \ensuremath{\pm} 2.67\\
Water-soluble extract content &
  56.85 \ensuremath{\pm} 0.97\\
Level of a soluble extract of ethanol &
  0.60 \ensuremath{\pm} 0.20\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\begin{table}[!htbp]
\caption{\boldmath {Calculation of standard solution for saponification} }
\label{tw-e7993976b941}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.23740000000000002\linewidth-2\tabcolsep}p{\dimexpr.17229999999999997\linewidth-2\tabcolsep}p{\dimexpr.1903\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}m KHP (99.5\%); Mr = 204 & NaOH volume & NaOH volume & HCl volume & N HCl\\
\rowcolor{kwdboxcolor} & (Vt\ensuremath{_{1}} ) & (Vt\ensuremath{_{2}} ) & (V\ensuremath{_{HCl}} ) & {\multirow{2}{*}{\cAlignHack (mol / L)}}\\
\rowcolor{kwdboxcolor}(gram) & (mL) & (mL) & (mL) & \\
\tblmidrule 
1.0165 &
  10.30 &
  10.45 &
  10.0 &
  0.5025\\
1.0170 &
  10.30 &
  10.30 &
  10.0 &
  0.4955\\
 &
   &
  \multicolumn{2}{p{\dimexpr(.3903\linewidth-2\tabcolsep)}}{\rAlignHack  Average N HCl } &
  0.4990\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\begin{table}[!htbp]
\caption{\boldmath {Calculation of saponification value} }
\label{tw-65e6791cb4d9}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1842\linewidth-2\tabcolsep}p{\dimexpr.25519999999999996\linewidth-2\tabcolsep}p{\dimexpr.26730000000000008\linewidth-2\tabcolsep}p{\dimexpr.2933\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}m Sample & Titration Blank & Sample titration & Saponification Number\\
\rowcolor{kwdboxcolor}(gram) & (mL) & (mL) & \\
\tblmidrule 
1.5205 &
  19.80 &
  16.75 &
  56.15\\
1.5519 &
  19.80 &
  16.70 &
  55.92\\
 &
   &
  \rAlignHack  Average &
  56.00\\
 &
   &
  \rAlignHack   x &
  0.23\\
 &
   &
  \rAlignHack  RPD &
  0.40\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}
\textbf{Saponification }\textbf{Value } \textbf{Test}

The results of the saponification value are as follows in Table~\ref{tw-e7993976b941} and Table~\ref{tw-65e6791cb4d9}. The saponification value was 56.00. 

\textbf{Characterization of fatty acid content}

Characterization of the kopyor extract has been carried out and showed that in the derivatization process, the extract was not stable, produced a blackish-brown liquid. Therefore the process of identification with MS GC was not carried out. 

\textbf{Minimum Inhibitory Concentration (MIC)} \textbf{ test}

\textit{MIC test on Staphylococcus aureus}

MIC test of standard against \textit{Staphylococcus aureus}, as shown in Table~\ref{tw-c46088f310db}.


\begin{table}[!htbp]
\caption{\boldmath {The MIC of standard against Staphylococcus aureus} }
\label{tw-c46088f310db}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4394\linewidth-2\tabcolsep}p{\dimexpr.5606\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Concentration (ppm) & Inhibition Zone Diameter (mm)\\
\rowcolor{kwdboxcolor} & Average \ensuremath{\pm} SD\\
\tblmidrule 
5.00 &
  19.03 \ensuremath{\pm} 1.89 \\
2.75 &
  11.40 \ensuremath{\pm} 0.54\\
2.50 &
  10.73 \ensuremath{\pm} 0.71\\
2.25 &
  9.42 \ensuremath{\pm} 1.51 \\
2.00 &
  8.65 \ensuremath{\pm} 2.19 \\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\begin{table}[!htbp]
\caption{\boldmath {MIC of standard against Pseudomonas aeruginosa} }
\label{tw-da415245e90c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4585999999999999\linewidth-2\tabcolsep}p{\dimexpr.5414000000000001\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Concentration (ppm) & Inhibition Zone Diameter (mm)\\
\rowcolor{kwdboxcolor} & Average \ensuremath{\pm} SD\\
\tblmidrule 
5.00 &
  15.70 \ensuremath{\pm} 0.96\\
2.75 &
  9.23 \ensuremath{\pm} 0.95 \\
2.50 &
  8.10 \ensuremath{\pm} 0.22\\
2.25 &
  -\\
2.00 &
  -\\
\tblbottomrule 
\end{tabulary}\par 
\end{table}

\begin{table}[!htbp]
\caption{\boldmath {The diameter of the inhibition zone of extract sand powders against Staphylococcus aureus and Pseudomonas aeruginosa} }
\label{tw-774e48e4ad4e}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3006\linewidth-2\tabcolsep}p{\dimexpr.3594\linewidth-2\tabcolsep}p{\dimexpr.34\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Samples   & \multicolumn{2}{p{\dimexpr(.6994\linewidth-2\tabcolsep)}}{\cAlignHack Inhibition Zone Diameter (mm) }\\
\rowcolor{kwdboxcolor} & Staphylococcus aureus  & Pseudomonas aeruginosa \\
\tblmidrule 
\multicolumn{3}{p{\dimexpr(1\linewidth-2\tabcolsep)}}{Cocos nucifera powder }\\
Positive Control &
  12.68  &
  10.65 \\
Powder sample 1 &
  14.22  &
  12.25 \\
Powder sample 2 &
  13.37  &
  11.32 \\
Powder sample 3 &
  14.17  &
  12.28 \\
Average &
  13.92   &
  11.95  \\
 &
   &
  \\
\multicolumn{3}{p{\dimexpr(1\linewidth-2\tabcolsep)}}{Cocos nucifera extract }\\
Positive Control &
  12.14  &
  10.88 \\
Extract sample 1 &
  14.25  &
  11.45 \\
Extract sample 2 &
  12.38  &
  12.21 \\
Extract sample 3 &
  13.30  &
  13.27 \\
Average &
  13.31   &
  12.31  \\
\tblbottomrule 
\end{tabulary}\par 
\end{table}
\textit{MIC test on }\textit{Pseudomonas aeruginosa} \textit{\space }

MIC test of standard against \textit{Pseudomonas aeruginosa}, as shown in Table~\ref{tw-da415245e90c}.

\textbf{Anti}\textbf{bacterial}\textbf{ activity test on }\textbf{Staphylococcus aureus,}\textbf{Pseudomonas aeruginosa} \textbf{and Staphylococcus epidermidis}

The activity results against Staphylococcus aureus and Pseudomonas aeruginosa were presented in Table~\ref{tw-774e48e4ad4e}.

The zone inhibition diameter of extracts and powder samples against the Staphylococcus aureus and Pseudomonas aeruginosa both showed equal activity and significantly showed higher activity compared to the standard solution. This showed the initial information that the powder and extract both have potential as antimicrobials. Antimicrobial activity can be applied to the skin or other route administration. To further strengthen the results for skin diseases, the activity of extracts and powder to the skin or topical disease was then conducted using skin bacteria such as Staphylococcus epidermidis. 

\textbf{Anti}\textbf{bacterial activity against} \textbf{ Staphylococcus epidermidis }

The antibacterial activity test against Staphylococcus epidermidis bacteria was shown in Table~\ref{tw-8dbfbb87e28b}. 


\begin{table}[!htbp]
\caption{\boldmath {Antibacterial activity against Staphylococcus epidermidis} }
\label{tw-8dbfbb87e28b}
\centering 
\begin{threeparttable}

\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2189\linewidth-2\tabcolsep}p{\dimexpr.1911\linewidth-2\tabcolsep}p{\dimexpr.3054\linewidth-2\tabcolsep}p{\dimexpr.2846\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sample  & Replication & \multicolumn{2}{p{\dimexpr(.59\linewidth-2\tabcolsep)}}{\cAlignHack Staphylococcus epidermidis }\\
\rowcolor{kwdboxcolor} &  & Inhibition Zone Diameter (mm) & Average (mm)\\
\tblmidrule 
 &
  1 &
  8.90 &
  \\
\cAlignHack F1 &
  2 &
  9.65 &
  9.45 \ensuremath{\pm} 0.48\\
 &
  3 &
  9.80 &
  \\
 &
  1 &
  6.50 &
  \\
\cAlignHack F2 &
  2 &
  6.50 &
  7.57 \ensuremath{\pm} 1.85\\
 &
  3 &
  9.70 &
  \\
 &
  Control (-) &
  5.50 &
  \\
 &
  Control (+) &
  6.60 &
  \\
\tblbottomrule 
\end{tabulary}\par 
\begin{tablenotes}\footnotesize 
    
\item{F1: Cocos nuciferaExtract; F2: Cocos nucifera Powder}
\end{tablenotes}
\end{threeparttable}

\end{table}
The diameter of the inhibitory zone against Staphylococcus epidermidis bacteria showed higher activity results compared to the standard solution. The kopyor Cocos Nucifera L. extract and powder demonstrated that both had potential as antimicrobials, especially to skin diseases. 
    
\section{Conclusion }
Cocos nucifera meat extract has been characterized compared to dried powder Cocos nucifera. The antibacterial activity results showed the potential of cocos Nucifera extract against \textit{Staphylococcus aureus, Pseudomonas aeruginosa and Staphylococcus epidermidis} bacteria\textit{.} This potential active agent can be suggested for further evaluation for topical disease application or other route administration.
\section*{Acknowledgement}The authors are very thankful to Universitas Airlanggafor providing the research grant and also thank the Faculty of Pharmacy Airlangga University (UNAIR) for supporting research facilities.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
